



Article Secure Decentralized IoT Service Platform Using Consortium Blockchain

Ruipeng Zhang D, Chen Xu and Mengjun Xie *D

Department of Computer Science and Engineering, The University of Tennessee at Chattanooga, Chattanooga, TN 37403, USA

* Correspondence: mengjun-xie@utc.edu

Abstract: Although many studies have been devoted to integrating blockchain into IoT device management, access control, data integrity, security, and privacy, blockchain-facilitated IoT communication is still much less studied. Blockchain has great potential in decentralizing and securing IoT communications. In this paper, we propose an innovative IoT service platform powered by the consortium blockchain technology. The proposed platform abstracts machine-to-machine (M2M) and human-to-machine (H2M) communications into services provided by IoT devices. Then, it materializes the data exchange of the IoT network through smart contracts and blockchain transactions. Additionally, we introduce the auxiliary storage layer to the proposed platform to address various off-chain data storage needs. Our proof-of-concept implementation was tested against various workloads and connection sizes under different block configurations to evaluate the platform's transaction throughput, latency, and hardware utilization. The experimental results demonstrate that our solution can maintain high performance with a throughput of approximately 800 reads per second (RPS), 50-80 transactions per second (TPS), and a latency of 50 ms-2 s under light to moderate workloads. Our extensive evaluation of the performance impact of batch size, batch timeout, and connection size also provides valuable insights into the optimization of blockchain configuration for achieving high performance.

Keywords: Internet of Things; IoT communication; security and privacy; consortium blockchain; smart contract; Hyperledger fabric

1. Introduction

Integration of the IoT and blockchain began to bloom thanks to distributed ledger technologies and cryptocurrencies. Previous IoT systems depend on centralized servers for communication and data storage, which often become the single point of risk to the security and privacy of the systems. Blockchain technologies, however, enable collaboration between untrusted parties in a decentralized manner. They eliminate the need for a trusted intermediary by creating a self-organized transaction network guided by a consensus protocol. Thanks to the distributed network architecture, data on blockchains not only have high availability but also strong integrity assured by cryptography algorithms and the immutable data structure. As a result, blockchain technologies are widely applied to fields of IoT, such as supply chains [1], power grids [2], healthcare [3], and smart homes [4].

However, prominent blockchain solutions have their performance, security, and privacy concerns. First, public permissionless blockchains, such as Bitcoin and Ethereum, oblige no restrictions on the participants who can create blocks and read transactions [5]. In the context of IoT, the openness of such blockchains endangers user privacy and exposes IoT systems to cyber attacks. In addition, the anonymity of permissionless blockchains makes it challenging to audit operations and trace in the network. Second, to reach global state consistency in a trustless environment, public permissionless blockchains use costly consensus protocols, such as proof of work (PoW), proof of stake (PoS), or protocols that



Citation: Zhang, R.; Xu, C.; Xie, M. Secure Decentralized IoT Service Platform Using Consortium Blockchain. *Sensors* 2022, 22, 8186. https://doi.org/10.3390/s22218186

Academic Editor: Ju Wook Jang

Received: 24 September 2022 Accepted: 20 October 2022 Published: 26 October 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). require particular hardware (e.g., proof of elapsed time) [6]. Such protocols are often not suitable for IoT systems where devices are heterogeneous and power-constrained. It is also difficult for them to meet the throughput and latency requirements of IoT applications, which often demand hundreds of transactions to be committed to the ledger within milliseconds to seconds.

Consortium blockchains, in comparison, remedy those disadvantages of public permissionless blockchains in a semi-trusted environment. Unlike permissionless blockchains, a consortium blockchain is operated by a group of collaborating entities, and only authorized nodes of these entities can commit blocks to the ledger [7]. On a consortium blockchain, the ledger can also be made visible to all members or part of the group via access control policies. Since participating authorities manage blockchain actors, it is more feasible to trace transaction flows and construct audit trails. Regarding consensus protocols, consortium blockchains assume that transaction validators are predefined and semi-trusted. Less resource-demanding consensus protocols, such as practical byzantine fault tolerance (PBFT) [8] and raft [9] can be utilized in consortium blockchains to improve scalability. Therefore, consortium blockchains usually yield much better performance than public permissionless blockchains [10].

1.1. Motivation

We reviewed existing IoT and blockchain integration in literature and discovered a research gap in realizing decentralized secure and scalable M2M and H2M communication with consortium blockchain technology. That is, overcoming the downsides of public permissionless blockchains for IoT communications by replacing them with consortium blockchains. A few research groups [11–13] explored the application of permissioned consortium blockchains and smart contracts in securing IoT communications and sensor data. However, they are either highly focused on a specific blockchain application of IoT, lack a robust system design, or miss an extensive evaluation of their proposed approach. There is a strong need for a more generalized, clearly defined, and extensively evaluated framework for IoT communications utilizing the consortium blockchain technology.

1.2. Contribution

In this paper, we present a <u>d</u>ecentralized <u>IoT</u> <u>service</u> <u>platform</u>, called DISP, for secure M2M and M2H communications inside an IoT environment based on a permissioned consortium blockchain. Instead of implementing a specific application of IoT with blockchain, the proposed work aims to establish a generic communication framework for various IoT systems. The consortium blockchain is a secure and scalable communication channel for IoT devices and applications in our solution. The communication protocol is formulated as services defined by IoT devices and provided to applications. Meanwhile, exchanging messages become blockchain transactions and are conveyed through the blockchain network. In addition, optional auxiliary storage is introduced to fit the proposed framework into diverse application scenarios, such as sensor data archives and real-time messaging. Finally, the framework provides a lightweight software development kit (SDK) and platform gateways to simplify blockchain operations for resource-constrained IoT devices and application developers. Compared to related studies, the proposed framework DISP has the following advantages:

(1) *Generality and versatility*: DISP is designed to power a wide range of IoT applications. It abstracts the communication protocol into the services that IoT devices can customize. It also imposes minimum assumptions about the underlying blockchain features or storage types. For example, it does not rely on a specific function offered by a particular blockchain platform or use a dedicated storage solution exclusively. Therefore, it can support various IoT applications to satisfy communication and data processing requirements.

(2) *Interoperability*: DISP can be integrated into existing IoT systems smoothly. With the help of straightforward SDKs and platform gateways, an application developer can not only bring new IoT devices to the proposed framework but also migrate legacy IoT devices

or systems easily. Meanwhile, DISP works well with existing IoT identity management and access control infrastructure and can reuse digital identities already in place.

(3) *High performance*: As illustrated by the performance evaluation results, our framework has considerable read and write throughputs and reasonable latencies even under high workloads. DISP works much more efficiently for semi-trusted consortium environments than permissionless blockchain-based IoT communication solutions, which rely on heavy consensus protocols.

Our contributions to this work are as follows:

(1) We propose a novel consortium blockchain-based IoT platform, namely DISP, for secure and decentralized IoT communications. DISP models IoT communications as services powered by smart contracts and blockchain transactions. In this paper, we elaborate on the system design and implementation that make DISP capable of supporting diverse IoT applications.

(2) We evaluate DISP extensively to demonstrate its performance under various workloads and block configurations. The experimental results illustrate that DISP can achieve high read and write throughputs and reasonable latencies for most workloads. This work also discusses how DISP can address security and privacy concerns.

(3) Two use-case studies, Parrot and Crystal Ball, are presented to showcase the versatility of DISP. Additionally, the DISP project—including its SDKs, use case demos, and testbed set-up scripts—is an open-source project to promote the reproducibility of this research.

1.3. Paper Organization

The remainder of this paper is organized as follows: Section 2 reviews related work of blockchain-based IoT identity management and access control, data storage and marketplace, and device manipulation. Section 3 describes the proposed IoT service platform's architecture and key processes. Section 4 briefly presents the proof of concept platform implementation and several use-case studies. Then, we evaluate the performance of the proposed platform with a series of experiments and discuss its security and privacy implications in Section 5. Finally, Section 6 concludes this paper with future research directions.

2. Related Work

The integration of blockchain and IoT has been extensively studied since the emergence of blockchain technology. Attempts have been made to address the challenges of IoT, such as distributed and heterogeneous architecture, device, data security, and profitability utilizing blockchains [14]. In this section, we will review the current development of IoT blockchain integration in three specific areas related to our proposed service platform: identity management and access control, data storage and marketplace, and device command and control.

2.1. Identity Management and Access Control

Identity management (IdM) and access control are keys to security and trust of IoT devices and data. It is difficult to apply traditional IdM systems to IoT environments, especially distributed and collaborative ones, due to their centralized nature, security vulnerabilities, and service fragmentation [15]. IoT's security, scalability, and interoperability requirements call for new IdM and access control paradigms, and blockchain-based solutions are promising answers.

One approach to building a blockchain-based IdM is recreating public key infrastructure (PKI) using blockchain and smart contracts [16–19]. A blockchain-based PKI supports the same critical operations, such as registration, verification, and revocation as a traditional centralized PKI, with improvements in security and privacy. Other research takes a different approach by building identity systems tailored to specific blockchain implementation. For instance, Sovrin [20] and its underlying Hyperledger Indy blockchain [21] provide a full-stack solution to decentralized, self-sovereign IdM on a public permissioned blockchain. Finally, storing identities off-chain (e.g., in traditional PKI) and linking them back to the blockchain were also discussed and utilized in projects, such as Hyperledger Fabric [22].

Once identity management has been established for an IoT system, one can further introduce access control to IoT data and regulate the communications between devices. Traditional access control methods, including role-based access control (RBAC), attribute-based access control (ABAC), and capability-based access control (CBAC) are less capable of supporting the enormous, heterogeneous, and decentralized IoT environments. Considering how access decisions are made, existing blockchain-based access control methods can be categorized into (1) transaction-based access control and (2) smart contract-based access control [23]. Transaction-based access control methods such as FairAccess [24] leverage blockchain as an immutable, distributed storage for access tokens, while the generation and verification of those tokens take place off-chain. On the other hand, smart contract-based access control focuses more on decentralizing the decision-making process with smart contracts. For example, IoTChain [25] allows resource owners to define smart contracts for granting client access and generating access tokens. IoT-CCAC [26] presents a consortium blockchain-based CBAC approach for IoT applications that exchange data between different consortium members.

2.2. Data Storage and Marketplace

The rapid advancement in distributed ledger technology invites new opportunities for distributed data storage, data sharing, and data monetization. Blockchain is a distributed, immutable database system where IoT data, such as sensor readings and access logs, may be stored. However, due to the block size limitation and scalability considerations, storing IoT data off-chain is more practical. Thus, hybrid blockchain-based storage networks have been proposed to reduce the chances of a single point of failure (SPOF) as seen in traditional centralized storage systems, provide data integrity and security, and lower the storage cost [27].

For instance, general-purpose blockchain-based solutions that support bulk data storage, such as Storj [28], Sia [29], and FileCoin [30], store only metadata of data blocks on the blockchain to prevent the bloating issue. On such platforms, files are split into smaller blocks and sent to the underlying distributed storage network composed of miners or storage nodes. To encourage participation and prevent dishonest behavior and free-riding, they also introduce new consensus protocols and cryptocurrencies to compensate for miners' storage and bandwidth usage.

The rise of blockchain technology and IoT also accelerates the growth of the market of IoT data. With blockchain technology, an IoT data marketplace can become fully decentralized and autonomous, while reducing cost, improving transaction efficiency, and promoting data privacy. Research in this area has been centered around ensuring data authenticity and provenance, secure data transfer, and payment processing [31–34]. Additionally, industry-led initiatives such as IOTA [35], XBR [36], and Streamr [37] also provide real-world insights into the monetization of IoT data on the blockchain.

2.3. Device Manipulation

Blockchain technology also enables fully decentralized M2M and H2M communications for IoT systems. With smart contracts, a blockchain-based IoT system is capable of autonomous decision-making based on business logic [14]. Such systems are relieved from centralized device management and control and, hence, less vulnerable to a SPOF. Meanwhile, the blockchain and smart contract ensure communication integrity and provide an immutable audit trail.

Slock.it [38] is among the first applications that leverage blockchain and smart contracts for controlling embedded devices. It envisages a blockchain-based economy of things where people can rent their unused assets, such as bikes, to others through smart contracts. A user can discover and lease assets on the Slock.it platform, and the whole process is enabled by

smart asset controllers or IoT devices connected to the Ethereum network. Upon successful payment, an asset will be unlocked by its controller as instructed by smart contracts.

Apart from securely sharing physical assets, Ethereum has been used as a decentralized M2M channel for IoT, thanks to its popularity and versatility. Fakhri and Mutijarsa [39] proposed a proof-of-concept demonstration on replacing MQTT [40] with the Ethereum blockchain for communications between IoT devices. The devices can talk to each other by reading data from and writing data to the blockchain via a smart contract intermediary. Wickström et al. [41] introduced an Ethereum-based protocol for IoT device management and task handling. This protocol utilizes two smart contracts to register an IoT device and create tasks for it. The authors pointed out that their protocol could reduce network attack risks to IoT devices because they can use the blockchain as a secure channel for remote command and control while disallowing other incoming network connections.

Finally, several studies have implemented decentralized IoT device management and communication using permissioned blockchain framework. Compared to permissionless blockchains, a permissioned blockchain inherently integrates with identity management and authentication so that the participation of the consensus can be verified and authorized. It usually comes with better scalability and energy efficiency because its consensus can be achieved without computationally expensive mining. Ali et al. [11] discussed a Hyperledger fabric-based IoT architecture for a smart home scenario where every device stores and shares its data via blockchain transactions. In such an environment, a smart device can request services from other devices by communicating directly or indirectly through the cloud. A smart contract also guards the list of devices and their shared secret keys. Hang and Kim [12] outlined a blockchain platform for securing IoT sensing data integrity. The platform provides smart contracts for registering and querying IoT devices and creating and deploying tasks that IoT devices can process. In the end, a device owner can receive notifications about the events generated by the tasks from the blockchain. The authors also detailed the implementation of their proposed platform that utilizes Hyperledger Fabric. Zhang et al. [13] applied Hyperledger Fabric to facilitate secure device communications in the edge computing environment. However, device communications and manipulations on their proposed platform are implemented through tasks that are scheduled and executed outside the blockchain.

2.4. Comparison of DISP and Related Work

A comparison of the proposed platform DISP with the aforementioned related studies is presented in Table 1. The research domain, research goal, applied blockchain technology, access control availability, and data storage mechanism are considered in the comparison. Distinct from other research, DISP focuses on securing the communication between IoT devices and their users. To realize this goal, we introduce authentication, access control, and data storage capabilities to DISP. DISP can be integrated into a broad spectrum of IoT scenarios rather than just a specific area of IoT. Moreover, DISP is the only one that explicitly considers the multi-organizational environment among all the compared studies. DISP also outperforms most related work in terms of system throughput and latency. The detailed performance comparison is given in Section 5.

Tabl	le 1.	Comparison	of	DISP	to 1	related	researc	h.
------	-------	------------	----	------	------	---------	---------	----

Related Work	Domain	Goal	Blockchain	Access Control	Data Storage
[13]	Edge computing	Access control	Hyperledger Fabric	Permissioned	Off-chain
[19]	General IoT	Access control	Hyperledger Fabric	Permissioned	On-chain
[26]	General IoT	Access control	BigchainDB	Permissioned	On-chain
[12]	General IoT	Data management	Hyperledger Fabric	Permissioned	On-chain
[39]	General IoT	Communication	Ethereum	Permissionless	On-chain
[41]	General IoT	Communication	Ethereum	Permissionless	On and off-chain
[11]	General IoT	Communication	Hyperledger Fabric	Permissioned	On-chain
Our work DISP	General IoT	Communication	Hyperledger Fabric	Permissioned	On and off-chain

3. System Design

An architectural overview of our proposed IoT service platform i.e., DISP, is illustrated in Figure 1. This overview shows the platform paradigm for a consortium composed of two organizations for demonstration purposes. At its core, the platform is powered by a consortium blockchain. Peer nodes from two organizations together serve the distributed ledger and smart contracts. These nodes also execute smart contracts when requested by the blockchain actors, which can be IoT devices or applications, and endorse blockchain transactions. IoT devices from any organization within the consortium may connect to the platform directly or through the platform gateway that handles blockchain operations on the device's behalf. To facilitate the integration of various IoT devices and accelerate application development, we also incorporate an SDK to reduce the complexity of devices and applications to interact with the platform. At least one identity service is required to provide digital identities to each IoT device, application, and peer node in every organization of the consortium, although multiple organizations may share the same identity service. Finally, auxiliary storage is introduced to enable data sharing between different organizations. The actual storage types and implementations are affected by individual application requirements.



Figure 1. Consortium blockchain-based IoT service platform architecture.

The following are the core components of DISP:

(1) *IoT devices*: IoT devices connect the physical space to cyberspace. Regarding DISP, they are the primary service providers. Services provided by IoT devices either measure or affect their environment, given that the device is a sensor or actuator. For example, a thermometer service reports the temperature of its surroundings, while a thermostat provides the service of adjusting room temperature. An IoT device may connect to the blockchain directly or through a platform gateway if its computational resources are constrained.

(2) *Platform gateway*: A platform gateway bridges the communication between IoT devices and the platform. Often, an IoT device is not able to participate in blockchain transactions directly due to energy or computational power constraints or because it is not programmable. In this case, it can delegate blockchain operations to the platform gateway without changing its inherent communication protocols.

(3) *Consortium blockchain*: At the core of the proposed platform, a consortium blockchain is employed as the distributed ledger that records all IoT devices and their services. It is also the primary communication channel between IoT services and their consumers, offering better performance and scalability compared to permission-less public blockchains. The IoT

network can become decentralized, meaning there are no longer centralized servers that are often SPOF. Moreover, the platform can leverage access control features introduced by consortium blockchains to secure IoT services from unauthorized access. Finally, since all changes to the IoT services and the service request and responses are recorded immutably on the ledger, the blockchain can essentially serve as a data historian for data auditing.

(4) *Peer*: Peers, or peer nodes, are computers that perform blockchain operations or offer auxiliary storage to IoT devices and applications of the platform. When serving as blockchain peers, they are responsible for hosting the distributed ledger and executing smart contracts. They are also essential in transaction endorsement. As for the auxiliary storage, a peer can be tailored to specific application needs. For instance, it can be a distributed network storage node, a distributed message queue broker, or a proxy server of media streams. Any organization in the consortium can contribute peers to the network. It is crucial to ensure a fair amount of peers from different organizations to achieve meaningful decentralization, and peers should be placed close to their neighboring IoT devices and gateways to reduce network overhead.

(5) Service registry and service broker: Service registry and broker are smart contracts that manage IoT devices and services add process service requests and responses. During device and service registration, the service registry updates the ledger with the latest device or service information provided by the IoT device. The service registry also provides interfaces for querying IoT devices and services. When invoked, the service broker smart contract will append service requests and responses to the ledger. IoT devices and applications observe and respond to ledger updates, eventually achieving asynchronous communications between service providers and consumers.

(6) *Platform SDK*: Platform SDK provides an application programming interface (API) of the platform to IoT service and application developers. It encapsulates functions for registering devices and services, sending service requests and responses, querying platform data, etc. The goal of platform SDK is to conceal the complexity of blockchain operations from IoT devices and application developers.

(7) *Application*: An application interacts with IoT devices via the services published on the platform. For example, an application can be an industrial control system (ICS) that monitors sensor readings or a smartphone app that displays room temperature and security camera feed in a smart home environment. Additionally, applications can provide services to other IoT devices and applications on the platform.

(8) *Identity service*: Every participating actor of the proposed platform, such as an IoT device, an application, or a peer node, is recognized by its digital identity. Therefore, we need an identity service for every organization to issue, renew, and revoke those identities. The platform allows organizations of the consortium to employ their own identity services so that each organization has complete control over its assets, such as peers, IoT devices, and applications. In addition, the consortium blockchain can enforce access control to IoT services through a set of policies defined for digital identities.

(9) *Auxiliary storage*: Although IoT devices and applications mainly communicate through the blockchain, it is sometimes desirable for IoT devices to share data off-chain. The proposed platform encompasses an optional auxiliary storage system to satisfy various data storage needs. For example, the auxiliary storage can be a distributed object storage system that stores historical humidity values, a distributed message queue for sharing real-time data, such as Particulate Matter (PM) 2.5 readings, or a proxy for streaming real-time binary data, such as a camera feed. The uniform resource identifier (URI) can be passed to the data consumer via IoT services. Finally, the data may be encrypted using the data consumer's identity to provide confidentiality.

The following sections elaborate on the necessary procedures for the IoT service platform DISP and explain how DISP takes advantage of the consortium blockchain and auxiliary storage to secure IoT systems. Figure 2 depicts the life cycle of an IoT device and its service on DISP. To expose functions to the network, an IoT device first must register itself and its services on the platform. Before the registration process starts, the device obtains its digital identity from the identity service of its belonging organization as marked by step (1) in Figure 2. Then, it announces the device information and services by invoking the service registry smart contract, which is shown in step (2) in Figure 2. If successfully validated, the device and service information will be stored on the blockchain. The registration process is required because device and service information is crucial to the authentication and access control processes, as described in Section 3.2.



Figure 2. The life cycle of an IoT device and its service.

(1) *Device Registration*: The first step for an IoT device to be registered DISP is receiving its digital identity from the identity service provider of its organization. Unlike many other blockchain-based IoT platforms in the literature that employ custom device identity and registration processes, our platform DISP can reuse existing digital identities issued by the organization's PKI. Thus, organizations can follow standard procedures of signing and issuing identity certificates using an off-the-shelf certificate authority (CA). Not only can DISP benefit from the robust security of a PKI, but also participating organizations can integrate their IoT device management system easily with DISP. Consequently, IoT devices can be decommissioned by deregistering the service from the service registry and putting their identity certificates on the certificate revocation list (CRL).

Next, the IoT device can register itself by invoking the service registry with the required information. Algorithm 1 is an abstraction of the device registration process executed by the service registry smart contract. The most important fields that must be provided by an IoT device are its digital identity and the organization ID, which are used to locate a specific device on the platform. Other information, such as name, description, and last update time, are human-readable metadata for platform users. The service registry first generates the unique device ID from the digital identity. Then, it validates the caller's identity to ensure it has sufficient permission to update the blockchain and does not impersonate other devices. Finally, the registry will serialize the provided device information and add it to the blockchain using the organization ID and device ID as the key.

Algorithm 1	Registering	a device with	the service registry.
-------------	-------------	---------------	-----------------------

Require:

D. Global State of the blockenant.	<i>B</i> :	Global state of the blockchain.
------------------------------------	------------	---------------------------------

- *P*: Service registry smart contract caller.
- C_{dev} : Device's digital identity. Must be unique within an organization.
- *I*_{org}: ID of the organization to which the device belongs.
- *N_{dev}*: Nickname of the device.
- D_{dev} : A short summary of the device's function and usage.
- T_{dev} : The date and time when the last update is made to the device information. **Ensure:**
- Device information is saved or updated in the blockchain.
- 1: **if** $WRITE \notin GETCALLERPERMISSIONS(P)$ **then**
- 2: return error ▷ Abort if caller is not allowed to update the blockchain
 3: end if
- 4: $I_{dev} \leftarrow \text{GENERATEDEVICEID}(C_{dev}) \qquad \triangleright \text{ Generate device ID from its digital identity}$ 5: **if** $I_{dev} \neq \text{GETCONTRACTCALLERID}(P)$ **then**
- 6: return error ▷ Abort if caller impersonates another device7: end if
- 8: if $I_{org} \neq \text{GetCONTRACTCALLERORGID}(P)$ then
- 9: return error ▷ Abort if caller impersonates another organization 10: end if
- 11: $S_{dev} \leftarrow \text{SERIALIZE}(I_{dev}, I_{org}, N_{dev}, D_{dev}, T_{dev})$
- 12: $B[I_{org}, I_{dev}] \leftarrow S_{dev}$
- 13: **return** S_{dev}

(2) Service Registration: Registered IoT devices publish their services on the network via the service registry smart contract. Such a service can measure a room's humidity or setting a refrigerator's temperature. An IoT device may declare multiple services on the blockchain by repeating the registration process. Similar to device registration, an IoT device is required to provide service information to the service registry for each service. The service registration process is outlined in Algorithm 2. The service name, device ID, and organization ID identify a unique service in the network. Therefore, each service has a unique service name on a device. The service version number and last update time are useful fields for application developers and system auditors to keep track of service updates. The description field offers a summary of the service and its usage. Besides identity and permission validation, the service registry will also check if the provided device is already registered on the platform. If there is no issue, it will write the serialized service information to the blockchain with the organization ID, device ID, and service name as the key. In accordance with the device decommission process, the service registry also allows IoT devices to deregister their services by calling the corresponding function in the smart contract, as shown in step (5) in Figure 2.

Require:	
<i>B</i> : Global state of the	he blockchain.
<i>P</i> : Service registry	smart contract caller.
N_{srv} : Name of the serv	vice. Must be unique among all services of the same device.
I_{dev} : ID of the device	that provides the service.
<i>I</i> _{org} : ID of the organiz	zation to which the device belongs.
<i>V_{srv}</i> : Version number	of the service.
D_{srv} : A short summar	ry of the service's function and usage.
T_{srv} : The date and tim	ne when the last update is made to the service information.
Ensure:	
Service information is	saved or updated in the blockchain.
1: if <i>WRITE</i> ∉ GetCall	LERPERMISSIONS (P) then
2: return error	Abort if caller is not allowed to update the blockchain
3: end if	
4: if $I_{dev} \neq \text{GetContract}$	CTCALLERID(<i>P</i>) then
5: return error	Abort if caller impersonates another device
6: end if	
7: if $I_{org} \neq \text{GETCONTRAC}$	CTCALLERORGID(<i>P</i>) then
8: return error	Abort if caller impersonates another organization
9: end if	
10: if GETDEVICE(I_{org} , I_{det}	$v_v) = \emptyset$ then
11: return error	Abort if device is not registered
12: end if	
13: $S_{srv} \leftarrow \text{SERIALIZE}(N_{srv})$	$v, I_{dev}, I_{org}, V_{srv}, D_{srv}, T_{srv})$
14: $B[I_{org}, I_{dev}, N_{srv}] \leftarrow S_{sr}$	v

3.2. Authentication and Access Control

DISP enforces authentication and access control to prevent unauthorized access. As stated in Section 3.1, the platform's authentication and access control rely on digital identities issued by the identity service. All the platform actors, such as IoT devices and applications, must have at least one valid digital identity to interact with the blockchain. The smart contracts will verify the identity of calling actors to see if they have the necessary permissions to read or write the distributed ledger. To enable cross-organization operations, the actors can possess multiple digital identities issued by different identity services, as depicted in Figure 3. Each identity, including its certificate *cert* and private key *prikey*, is stored in a blockchain wallet W. Therefore, the platform allows an IoT device to register its service in multiple organizations using different identities stored in the same wallet, i.e., $W = \{(cert_i, prikey_i) | i = 1, 2, ..., N\}$.

DISP enforces more granular access control via a multi-layer RBAC model, as shown in Figure 3. In this model, the digital identity of each actor is assigned a role when it is created. The first layer, the blockchain-level access control policies, define which organizations and roles can query or update the ledger using smart contracts. By default, actors of a participating organization can see all registered IoT devices and services, but only the actors with the "writer" roles have the right to register or update them. At the next layer, the transaction level, service registry, and service broker smart contracts can be attached with transaction validation policies. These policies tell which organizations and actors must sign the service request transaction for it to be valid. Finally, the smart contracts enforce the last layer of access control for each IoT device, service, service request, and service response. The smart contracts check the caller's identity to ensure that only the service-owning IoT device can update the device and service information and respond to service requests.



Figure 3. The proposed multi-layered access control model.

3.3. Service Request and Response

The consortium blockchain provides opportunities for IoT networks to become decentralized and distributed. It also enables sharing of IoT infrastructure among organizations within a consortium in a scalable and reliable manner. Leveraging the advantages of consortium blockchain, we propose an IoT communication process that utilizes the blockchain network as the communication channel. In this process, data exchange between parties is achieved in service requests and responses facilitated by the service broker smart contract. Figure 4 provides a more detailed depiction of steps (3) and (4) of Figure 2. It illustrates the communication process between an IoT device and an application, including querying available services, requesting services, responding to service requests, and retrieving data.



Figure 4. Process of communication between an IoT device and application.

(1) Service querying: To find out a service's availability, an application can query the service registry before initiating a service request, as illustrated by step (1) in Figure 4. Given the service name, device ID, and organization, the service registry returns information about the service to the application, such as the service version and last update time. Although this step is not mandatory, an application is recommended to refresh service information periodically to stay updated on the service's information and look for alternative services when the service is unavailable. Additionally, the last update time of service helps decide the best time to refresh such information.

(2) *Service request and processing*: To communicate with an IoT device on DISP, an application creates a service request first. Every service request is identified by a universally unique identifier (UUID) generated by the application. The request ID must be unique among all requests on the platform. Information about the requested service, such as the service name, is required so that the service-providing IoT device can retrieve the request from the blockchain. The request also contains the request creation time used for request deduplication and auditing purposes. The request body is represented by the method and arguments fields where the action and its optional arguments are defined.

Next, the application submits the service request to the service broker, which will validate the request and create a blockchain transaction. Algorithm 3 illustrates the service request processing at a high level. The service broker smart contract first verifies the contract caller's permissions to ensure it can update the blockchain. Additionally, it validates the requested service and skips duplicated requests. The request is then serialized and saved in the blockchain through a blockchain transaction. Once the transaction is validated and endorsed by the network and written to the ledger, the service-providing IoT device will be notified by the network that its service has been requested. An IoT device can then fetch request information from the blockchain via the service broker and interact with its environment as instructed by the request. These processes are depicted in steps (2) and (3) in Figure 4.

Algorithm 3 Creating a service request using the service broker.	
Require	

Require:

- *B*: Global state of the blockchain.
- *P*: Service broker smart contract caller.
- *I_{req}*: UUID of the request.
- T_{req} : Date and time when the request is created.
- *N*_{srv}: Name of the requested service.
- I_{dev} : ID of the device that provides the service.
- *I*_{org}: ID of the organization to which the device belongs.
- M_{req} : Request method.
- A_{req} : Optional arguments of the request.

Ensure:

Service request information is saved in the blockchain.

- 1: **if** $WRITE \notin GETCALLERPERMISSIONS(P)$ **then**
- 2: return error ▷ Abort if caller is not allowed to update the blockchain
 3: end if
- 4: **if** GETSERVICE(I_{org} , I_{dev} , N_{srv}) = \emptyset **then**

return error

- return error > Abort if requested service is not registered
- 6: **end if**

5:

8:

- 7: **if** GETSERVICEREQUEST(I_{req}) $\neq \emptyset$ **then**
- Abort if service request already exists

- 9: end if
- 10: $S_{req} \leftarrow \text{SERIALIZE}(I_{req}, T_{req}, N_{srv}, I_{dev}, I_{org}, M_{req}, A_{req})$
- 11: $B[I_{req}] \leftarrow S_{req}$
- 12: NOTIFYEVENT (S_{req})
- 13: return S_{req}

(3) Service response and data retrieval: IoT devices can respond to service requests asynchronously whenever they finish processing them. The process of replying to a service request is shown in steps (4) to (6) in Figure 4. An IoT device starts by creating the service response. Besides the UUID of the corresponding request, the response also contains fields about response creation time, a custom status code indicating the status of processing, and an optional value to be returned to the requester. The device may also store operational data in the auxiliary storage and leave a pointer to the data in the return value under situations where the volume or time sensitivity of the data cannot be met by the blockchain. Details of the auxiliary storage are further discussed in Section 3.4.

Following its creation, the service response is sent by the IoT device to the service broker, which creates another blockchain transaction. Algorithm 4 depicts the service response creation process in the service broker. Similar to the service request creation, the broker validates the caller's identity and the existence of the request. It also prevents service impersonation by matching the caller's identity against the identity of the requested service. The service response update transaction will undergo the same process of validation and endorsement as the service request transaction and eventually be appended to the distributed ledger. By listening to transaction events on the blockchain, an application can act upon the completion of its request, e.g., retrieving the response from the blockchain. Furthermore, if the return value contains a pointer to data in the auxiliary storage, the application can perform additional storage operations to retrieve the data.

Require:	
B: Global state of the	blockchain.
<i>P</i> : Service broker small	art contract caller.
<i>I_{req}</i> : UUID of the reque	est to which the response answers.
T_{res} : Date and time whe	en the response is created.
U_{res} : The request proces	ssing status.
<i>R_{res}</i> : Optional value to	be returned to the service request sender.
Ensure:	
Service request informat	ion is saved in the blockchain.
1: if $WRITE \notin GETCALLE$	RPERMISSIONS (P) then
2: return error	Abort if caller is not allowed to update the blockchain
3: end if	
4: if GETSERVICE(<i>I</i> _{org} , <i>I</i> _{dev} ,	$N_{srv}) = \emptyset$ then
5: return error	Abort if requested service is not registered
6: end if	
7: $S_{req} = \text{GetServiceReq}$	$UEST(I_{req})$
8: if $S_{req} = \emptyset$ then	
9: return error	Abort if service request does not exist
10: end if	
11: if GetDeviceId(S_{req}) \neq	\notin GETCONTRACTCALLERID(P) then
12: return error	Abort if caller impersonates another device
13: end if	
14: if GetOrganizationII	$D(S_{req}) \neq \text{GETCONTRACTCALLERORGID}(P)$ then
15: return error	Abort if caller impersonates another organization
16: end if	
17: $S_{res} \leftarrow \text{SERIALIZE}(I_{req}, T)$	$T_{res}, U_{res}, R_{res})$
18: $B[I_{req}] \leftarrow \{S_{req}, S_{res}\}$	
19: NOTIFYEVENT (S_{res})	
20: return S_{res}	

3.4. Data Storage

Due to blockchain's transaction size, throughput, and latency limitations, it is often preferable to store IoT-generated data off-chain. To meet the storage requirements of different IoT applications, auxiliary storage can be included to complement the data exchange on the platform. Figure 5 demonstrates how auxiliary storage facilitates IoT devices to pass the data to an application. The storage system can be composed of different types of data storage depending on the communication requirements. For example, nonreal-time structured data can be stored in a cloud, such as AWS S3, or a distributed object storage, such as the InterPlanetary File System (IPFS). IoT device logs and historical sensor readings fall into this category. Meanwhile, real-time sensor data (e.g., instantaneous geolocation coordinates) can be published to message queues such as Apache Kafka for efficient data delivery. Finally, media streaming servers can be included in the auxiliary storage to transcode, cache, and stream multimedia captured by IoT devices to their users. Examples of these data types are surveillance camera feed and audio data collected by an acoustic gunshot detection system. Each type of storage can be offered by a single organization of the consortium or hosted by multiple organizations collaboratively. The design of data storage is out of the scope of this paper.



Figure 5. Auxiliary storage types and applications.

4. Implementation and Case Studies

This section dives into the implementation details of the proposed platform. We further showcase the capabilities and usefulness of the platform with two real-world IoT applications. For the sake of research reproducibility, the source code of our implementation, the exemplary case study applications, and all the scripts for testbed setup and benchmarking are made available online (Please refer to the Data Availability Statement).

4.1. Platform Implementation

Considering the feature richness and development support, we selected Hyperledger Fabric [22] as the consortium blockchain platform for our IoT service platform implementation. Thanks to its modular architecture and plug-and-play nature, Hyperledger Fabric has been widely used in industrial environments. Our implementation takes advantage of Hyperledger Fabric's components to realize the core functionality of DISP in the following aspects: (1) We incorporated device and user identities into the blockchain using Hyperledger Fabric's membership service provider (MSP). Each organization on the platform has its dedicated MSP, which translates the identities into roles and privileges of the blockchain. Thus, the platform can authenticate invocations to smart contracts using registered identities.

(2) Hyperledger Fabric defines various policies agreed upon by the consortium members, or channel members in Hyperledger Fabric's terminology, for infrastructure management. In our implementation, we limit IoT devices and users to only submitting transactions or querying the ledger using ACL. We also restrict which organizations must approve or endorse the transactions with smart contract endorsement policies. In addition, smart contracts also limit access to write operations, such as responding to service requests to the device and service owners by checking the caller's identity.

(3) The smart contracts of the proposed platform were implemented using Fabric contract API in the Go programming language. All smart contracts are packaged in one chaincode, a container for smart contracts, so that they share the same world state. In addition, we created SDK for Go, Java, JavaScript, and TypeScript programming languages to simplify application development for DISP.

(4) Client communications with the blockchain are simplified using the Hyperledger Fabric gateway. Instead of directly interacting with the blockchain network and managing the complexity of transaction proposal, endorsement, and commission, IoT devices and users delegate most of the heavy-lifting operations to a gateway component running on peer nodes. This improvement is essential for devices that are energy-constrained or low on computing power.

4.2. Case Study: Parrot

Parrot is a voice assistant for the smart home lighting system implemented using the proposed IoT service blockchain. It enables touchless control of home lights using voice commands, such as "Parrot, turn on the kitchen light". The architecture of Parrot is shown in Figure 6. The workflow starts with a user issuing a voice command in a given format to a smart speaker. A microphone on the smart speaker continuously listens in the background but only begins recording voice commands on wake words, "Parrot" in this case. It also determines the duration of the recording and stops when the command ends. This is done using an onboard wake word detection engine, Porcupine [42], and the voice activity detector provided by WebRTC [43]. Then, the smart speaker adds the recording to a private decentralized file storage network, implemented by IPFS, and calls the service exposed by a remote voice AI engine to the IoT service platform. Next, the AI engine is notified by the blockchain and retrieves the user's audio recording from IPFS. The data are then fed into Rhino [44], a speech-to-intent engine that decodes voice commands and extracts the location of the light and actions to perform. Finally, the AI engine sends requests for turning on or off to the corresponding actuator, i.e., the smart light, via the blockchain.

The main advantage of this model is that all communications are securely backed by the proposed platform DISP. Compared to traditional IoT systems, actors of this system publish their services only to the blockchain instead of exposing them using other insecure communication protocols. Moreover, all processes involved are transparent to users since service activities are recorded on the immutable blockchain. Blockchain transactions leave an audit trail that is invaluable to incident response and forensic investigation when a problem arises. Regarding user privacy, the network operator can isolate different user groups using separate Hyperledger Fabric channels. Audio recordings of the user can also be set to expire automatically by unpinning and garbage collecting them from the storage network.



Figure 6. The architecture and data flow of Parrot.

4.3. Case Study: Crystal Ball

Surveillance cameras are widely deployed nowadays. However, they are often vulnerable to hackers or malware due to poor security design and improper configuration. An unprotected surveillance camera can seriously threaten user privacy as the video footage may be leaked to unauthorized parties. For example, Insecam (http://www.insecam.org (accessed on 22 September 2022)) is a live camera directory that allows visitors to view live streams from thousands of unprotected public cameras as of May 2022, and the number of exposed cameras is still growing. Hackers can also use a compromised camera in other cyber crimes, e.g., to form a botnet and initiate distributed denial-of-service (DDoS) attacks [45].

We have built a blockchain-based secure surveillance streaming system called Crystal Ball using DISP. Figure 7 depicts the architecture and workflow of Crystal Ball. A camera in the Crystal Ball system does not serve its video and audio feeds on an open network port. Instead, it publishes them to a secure streaming server that streams camera feeds only to users with correct access tokens. Meanwhile, the streaming server creates an IoT service that generates and distributes one-time session access tokens to authorized users on the blockchain. Finally, users can request access tokens and watch live streams from an Android streaming client.



Figure 7. The architecture and data flow of Crystal Ball.

Crystal Ball eliminates the need for exposing camera feeds from the capturing device. Unnecessary ports can now be closed to reduce the attack surface of the camera. Furthermore, Crystal Ball protects camera feeds from unauthorized access using device identities and access tokens. As access tokens are one-time only and tied to each session, a malicious user cannot reuse previous tokens even if they are leaked. Finally, an administrator can easily log and analyze camera accesses using transaction history and decommission cameras that are compromised by invalidating their device identities.

5. Evaluation and Discussion

In this section, we first present the experiment settings under which we evaluated the proposed IoT service platform DISP. Our evaluation focuses on two core configuration parameters of the blockchain, transaction batch size, and batch timeout, to discover how blockchain configuration affects the performance of DISP. We then examine the transaction performance overhead introduced by additional IoT device connections. Transaction throughput, transaction latency, and system resource utilization are measured in each test as the performance indicators of the platform. Finally, we discuss DISP's security and privacy and its impact on IoT systems.

5.1. Experimental Setup and Methodology

Figure 8 illustrates the architecture of our testbed and Table 2 details the hardware and software configuration of each machine used in our experiments. The proposed IoT service platform ran on a multi-node Hyperledger Fabric blockchain comprising two peer organizations and one orderer organization. Each peer organization contains two peer nodes. The orderer organization contains three Raft orderer nodes, and all run Hyperledger Fabric version v2.4.3. All the organizations form a single consortium, and transactions are executed and ordered in a single channel. The channel enforces the "MAJORITY" endorsement policy, which means the two peer organizations must both endorse the transaction for it to be committed to the blockchain. Then, the IoT service blockchain chaincode is installed on all the peer nodes. To resemble a multi-organizational environment, we deployed the testing blockchain on bare-metal servers across two Chameleon Cloud [46] data centers. Compared to deploying all nodes to the same data center, our setup introduces a network latency of around 30 ms between consortium parties to reflect a network environment of a real-world blockchain network.

Node Name	CPU (Cores)	Memory (GB)	Disk (GB)	Network (Gbps)	Location
Hyperledger Fabric Orderer1	24	128	233	1	Chicago, IL, USA
Hyperledger Fabric Orderer2	24	128	233	1	Chicago, IL, USA
Hyperledger Fabric Orderer3	24	128	233	1	Chicago, IL, USA
Hyperledger Fabric Org1 Peer1	48	191	447	10	Austin, TX, USA
Hyperledger Fabric Org1 Peer2	48	191	447	10	Austin, TX, USA
Hyperledger Fabric Org2 Peer1	48	191	447	10	Austin, TX, USA
Hyperledger Fabric Org2 Peer2	48	191	447	10	Austin, TX, USA
Hyperledger Caliper Manager	2	2	40	1	Chattanooga, TN, USA
Hyperledger Caliper Worker1	2	2	40	1	Chattanooga, TN, USA
Hyperledger Caliper Worker2	2	2	40	1	Chattanooga, TN, USA
Hyperledger Caliper Worker3	2	2	40	1	Chattanooga, TN, USA
Hyperledger Caliper Worker4	2	2	40	1	Chattanooga, TN, USA
Hyperledger Caliper Worker5	2	2	40	1	Chattanooga, TN, USA

Table 2. Hardware configurations of Hyperledger Fabric nodes and Hyperledger Caliper nodes.



Figure 8. IoT service platform testbed architecture.

Apart from the Hyperledger Fabric blockchain nodes, six virtual machines were employed to establish the Hyperledger Caliper [47] benchmarking environment. Hyperledger Caliper is a blockchain benchmarking tool that generates synthetic transaction workloads and measures the performance of the system under test (SUT). Our benchmarking environment is composed of a Caliper manager and five Caliper workers. The manager distributes workload parameters to the workers, synchronizes workers during each test, collects evaluation results, and generates human-readable reports. Each of the five workers starts two client connections to the blockchain, which simulates ten IoT devices that provide or request services on the proposed platform. These workers execute the workload scripts that call the smart contracts and generate transactions guided by the fixed-load rate control strategy. This strategy guides the workers to send the transactions or queries at a dynamic rate such that the number of incomplete transactions or queries in the SUT always stays under a given value. In our experiments, the workers collectively send 2000 transactions or queries under a fixed load of 100 transactions/queries. Finally, updating the same device, service, service request, or response in a single test is avoided in order to work around MVCC_READ_CONFLICT errors in Hyperledger Fabric. Compared to the blockchains such as Ethereum that have a more serialized approach to transaction processing, Hyperledger Fabric employs lock-free optimistic concurrency. To prevent the double-spending problem, Hyperledger Fabric will only process one of all transactions that modify the same data and reject the rest in the same block [48].

The core metrics inspected in each test include peak read/transaction throughput and average read/transaction latency. According to the Hyperledger Blockchain Performance Metrics white paper [49], a read operation does not change the ledger state, while a transaction operation involves ledger updates. The latency L_{read} of a read operation measured in seconds is defined as:

$$L_{read} = t_{response} - t_{submit},\tag{1}$$

where t_{submit} is the read request submission time and $t_{response}$ is the time at which the reply is received. The throughput W_{read} of read operations measured in RPS is defined as:

$$W_{read} = \frac{N_{read}}{T_{read}},\tag{2}$$

where N_{read} is the total number of read operations completed in time T_{read} .

Transaction throughput and latency are measured differently from the read throughput and latency since the confirmation time of blocks must be considered in transaction operations. The transaction latency L_{tx} in seconds is defined as:

$$L_{tx} = t_{confirm} - t_{submit},\tag{3}$$

where $t_{confirm}$ is the time at which the transaction is confirmed by the network given a network threshold (e.g., 90% of the network), and t_{submit} is the time when the transaction is submitted by the client. The transaction throughput W_{tx} measured in TPS is defined as:

$$\mathcal{N}_{tx} = \frac{N_{tx}}{T_{tx}},\tag{4}$$

where N_{tx} is the total number of committed transactions at all nodes of the network in time T_{tx} .

System resource utilization metrics are also measured during each test. They include average CPU and memory usage, total data sent to or received from the network, and total data read from or written to the disk. These metrics are polled and aggregated by Prometheus [50] periodically from each Hyperledger Fabric node and reported to the Caliper manager. The read/transaction throughput and latency of the blockchain and the above resource utilization metrics collectively indicate how well our proposed platform performs under different blockchain configurations. They also provide valuable information on optimizing the blockchain for various use cases.

5.2. Performance vs. Batch Size

We first examine the impact of transaction batch size on the performance of DISP. Three parameters constrain the batch size in the Hyperledger Fabric blockchain: maximum message count N_{max} , preferred maximum bytes of messages S_{soft} , and absolute maximum bytes of messages S_{hard} . Therefore, the batch size configuration directly controls the number of messages N_{msg} that can be batched into a block. When the size of each message, or transaction data, is relatively small, N_{msg} is primarily limited by N_{max} . Otherwise, N_{msg} will mostly be limited by S_{soft} unless there are many huge messages whose sizes exceed S_{hard} .

In our experiments, we intend to control the batch size solely with S_{soft} . So we assign large constant values to N_{max} and S_{hard} to ensure blocks always reach S_{soft} first. Then, we set S_{soft} to 128 KB, 256 KB, 512 KB, 1 MB, 2 MB, 4 MB, and 8 MB, respectively, and evaluate the smart contract operations of DISP. Device registration and deregistration transaction throughput and latency are omitted from the results since the workers only start ten blockchain clients for the tests. As each IoT device binds to one client identity, the workers are limited to registering or deregistering ten devices in total. Therefore, insufficient transactions can be evaluated for the above two operations to report their accurate throughput and latency results, and we exclude these operations from the result analysis. However, we will revisit these operations and their performance in later experiments.

Figures 9 and 10 present the throughput and latency of read and transaction operations under various batch sizes. The throughput and latency of all read operations stay nearly unchanged as batch size increases. This is because read operations in Hyperledger Fabric are not sent to the ordering service for validation and committing to the ledger [51]. Thus, batch size configuration has no impact on the performance of read operations. On the other hand, the size of messages has more impact on the performance metrics. The "query all

requests" and "query all services" operations have lower throughput and higher latency values as the messages returned to the clients are significantly larger than the singular queries (e.g., "query a device") and device queries. It can take more time and bandwidth for the peers to process and transmit such messages, resulting in lower performance. In contrast, the performance metrics of transaction operations are tied more closely to batch sizes. The throughput and latency degrade quickly from 100 TPS and 0.4 s to 25 TPS and 2.4 s as the batch size increases until it reaches 1 MB since larger blocks need more time to generate and dispatch. Further increasing the batch size will not worsen the performance.





Figure 9. Throughput and latency of read operations with varying batch sizes.

Figure 10. Throughput and latency of transaction operations with varying batch sizes.

Before investigating batch size's impact on system resource utilization of smart contract operations, we first explore the similarities in resource utilization patterns of different operations. Figures 11 and 12 present the average CPU, average memory, total network, and total disk usage of every host of the test Hyperledger Fabric network with 2 MB batches. The results show that the peer nodes are more utilized than the orderer nodes under most workloads. Moreover, the utilization levels are even on the same type of nodes. Regardless of the specific operation performed, all the read operations show similar resource usage patterns, and so are the transaction operations. Meanwhile, the transaction operations involve more incoming network traffic and disk writes than the read operations, and the operations that consume and produce larger messages require more CPU and memory. Finally, disks are rarely read in all the tests due to ledger data being loaded to memory before the measurements begin.



Figure 11. The average CPU usage (**a**), average memory usage (**b**), total data received from the network (**c**) and sent to the network (**d**), total data written to disk (**e**) and read from disk (**f**) of orderer1 (O_1), orderer2 (O_2), orderer3 (O_3), org1 peer1 (P_{11}), org1 peer2 (P_{12}), org2 peer1 (P_{21}), and org2 peer2 (P_{22}) during read operations when $S_{soft} = 2$ MB.

The same resource utilization patterns are also observed in other tests with varying batch sizes. Therefore, we only present the results of the "request for a service" and "query a request" operations to demonstrate the relationship between resource utilization and batch size. Figures 13 and 14 depict the CPU, memory, network, and disk usage of each test given each batch size. For read operations such as querying a single service request, there is slight fluctuation in system resource usage as the batch size increases, similar to their throughput and latency. This result aligns with the fact that those operations do not go through the ordering process. On the other hand, transaction operations incur a higher CPU utilization and more disk writes when the batch size is small. It is clear that with smaller batch sizes, the network needs to create more blocks to commit the same number of transactions, hence, more overhead in the data to be written to disk and more computing power required to complete that. The network usage does not change significantly with batch size because the sizes of incoming and outgoing messages remain the same regardless of batch size.



Figure 12. The average CPU usage (**a**), average memory usage (**b**), total data received from the network (**c**) and sent to the network (**d**), total data written to disk (**e**) and read from disk (**f**) of orderer1 (O_1), orderer2 (O_2), orderer3 (O_3), org1 peer1 (P_{11}), org1 peer2 (P_{12}), org2 peer1 (P_{21}), and org2 peer2 (P_{22}) during transaction operations when $S_{soft} = 2$ MB.



Figure 13. The average CPU and memory usage (**a**), total data received from and sent to the network (**b**), total data written to and read from disk (**c**) of orderer and peer nodes for handling "querying a service request" operations with varying batch sizes.



Figure 14. The average CPU and memory usage (**a**), total data received from and sent to the network (**b**), total data written to and read from disk (**c**) of orderer and peer nodes for handling "requesting for service" operations with varying batch sizes.

5.3. Performance vs. Batch Timeout

Batch timeout ($T_{timeout}$) is another essential configuration that controls the generation of blocks. It is the maximum time to wait before creating a new block after the first transaction arrives at the ordering service. To estimate the impact of batch timeout on the performance of the proposed system, we perform the same experiments as the ones in the previous section, with batch timeout set to 500 ms, 1 s, 2 s, 4 s, and 8 s, respectively. The throughput and latency of the read and transaction operations are shown in Figure 15 and Figure 16, respectively. Similar to the results of batch size tests, the throughput and latency of read operations are unaffected by the change of batch timeout because the ordering service does not process such operations. Transaction operations show a logarithmic decrease in throughput and a linear increase in latency regarding batch timeout. Blocks wait longer to be created when timeout is large, and the commit of transactions to be batched into the same block, mitigating the decrease in throughput.



Figure 15. Throughput and latency of read operations for varying batch timeouts.

Figures 17 and 18 depict the average system resource utilization of orderer and peer nodes for the tested batch timeouts. Similar to the previous experiments, read operations exhibit consistent resource usage given different batch timeouts. Transaction operations, on the other hand, consume more orderer and peer CPU for small batch timeouts. The CPU usage decreases in the same fashion as the throughput when the timeout increases. A steady climb in the outgoing network traffic and disk write can also be observed as the



timeout increases. Other resources, such as memory and incoming network traffic, are much less affected by the batch timeout.

Figure 16. Throughput and latency of transaction operations for varying batch timeouts.



Figure 17. The average CPU and memory usage (**a**), total data received from and sent to the network (**b**), total data written to and read from disk (**c**) of orderer and peer nodes for handling "querying a service request" operations with varying batch timeouts.





5.4. Performance vs. Connection Size

A real-world IoT network can be composed of hundreds of devices that actively provide services simultaneously. The increase in device connections inevitably affects the responsiveness of our proposed platform. Therefore, our final evaluation focuses on the impact of client connection size on DISP's performance and system resource utilization. We created 2000 client connections to the platform using Hyperledger Caliper in order to simulate the situation where a massive number of IoT devices communicate through the platform concurrently. Then, we rerun the tests with the batch size configured to 2 MB and batch time to 2 s. The results are compared with the previous results where ten client connections were employed to discover the performance and resource utilization overhead of DISP with a massive number of connections

Figure 19 presents the throughput and latency of certain operations under two connection size settings. Batch queries such as "query all services" are excluded from the evaluation. This is because the numbers of devices or services they query are different under the two testing scenarios, which makes the results incomparable. For the read operations, the throughput experiences a 5% to 6% decrease in the 2000 connections scenario while the latency remains the same. For the transaction operations, the throughput suffers a 30% degradation, and the latency increases by 25% to 38%. These notable overheads result from massive clients collecting and submitting endorsements from peers. The more connection established to the peers and orderers, the more congestion there will be hindering transaction processing. Therefore, a recommendation for performance tuning would be to limit the number of client connections to each peer.



Figure 19. Throughput and latency of read and transaction operations for 10 client connections $(10 \times)$ and 2000 client connections $(2000 \times)$.

The system resource utilization also rises in the 2000-connection scenario, as shown in Figure 20. The most significant increases in relative resource utilization happen in networking, where a hundred-time growth in outgoing network traffic can be observed. A drastic usage increase in CPU and memory can also be observed for both read and transaction operations. Additionally, transaction operations are more fickle to changes in connection size compared to read operations, and so do peer nodes than orderer nodes. The disk usage, however, does not vary too much with the connection size. The overhead observed in system resource utilization could be attributed to the connection overhead as the nodes have to maintain the connections from each client, verify client identities, and secure mutual communications using encryption. The results also imply that the memory and network can quickly become the bottleneck of the blockchain platform for large IoT networks.



Figure 20. Average system resource utilization of orderer and peer nodes for 10 client connections $(10\times)$ and 2000 client connections $(200\times)$.

5.5. Performance Comparison with Related Work

We compare the evaluation setup and performance of our proposed platform with those in other related studies in the literature. The comparison results are shown in Table 3. Unfortunately, many related studies neither detail the settings for evaluation nor explain their measurements. Therefore, we mark the missing data as "Not available". For the studies that provide performance evaluation results, we capture the average throughputs and latencies for read and transaction operations. The performance results of some platforms were measured under different evaluation setups and obtained with different methods (e.g., round-trip time (RTT) instead of latency). Nevertheless, these results are valuable to evaluate related blockchain-based IoT platforms at a high level. Finally, we include our solution's average read and transaction throughputs and latencies, excluding the results of batch queries when the batch size is set to 2 MB and timeout set to 2 s.

Related Work	Network Size	Client Size	Read Throughput/Latency	Transaction Throughput/Latency
[13]	5 orderers + 10 peers	3	Not available	Not available/25–183 ms
[19]	Not available	Not available	7–20 RPS/59.5–69 ms	4–19 TPS/161–205.5 ms
[26]	1 BigchainDB server	Not available	Not available	Not available/640–1210 ms
[12]	Not available	50, 150, 250, 500	Not available/271–752 ms	Not available/2490–3012 ms
[39]	Not available	Not available	Not available	Not available
[41]	Not available	Not available	Not available	Not available
[11]	Not available	Not available	2750–3250 RPS/260–600 ms	2250–3250 TPS/300–700 ms
Our work DISP	3 orderers + 4 peers	10, 2000	700–800 RPS/50 ms	18–25 TPS/2000–2500 ms

Table 3. Performance comparison of the proposed platform to related work.

As it is shown in Table 3, our proposed platform DISP achieves much better throughput and latency for read operations than the other platforms while providing comparable throughput and latency for transaction operations. Our evaluation setup is also closer to a real-world distributed system setup than the others, as a geographically distributed consortium with multiple organizations was simulated. Moreover, the scalability of DISP was evaluated using different client or connection sizes, while most of the other studies do not provide such information. In summary, the extensive evaluation demonstrates a favorable performance result of our solution compared to the other solutions in the literature.

5.6. Security and Privacy

It is essential that our proposed platform DISP provides a secure environment for IoT devices and applications by leveraging blockchain technologies and modern cryptography. It is also important that we preserve user and data privacy wherever needed. In this section, we examine the security and privacy of DISP from the perspectives of blockchain, auxiliary storage, and IoT devices.

(1) *Blockchain security and privacy.* Ferrag et al. [52] presented a comprehensive review of the thread models and attacks against blockchain systems. They classified attacks on blockchain systems into five categories: identity-based attacks, manipulation-based attacks, cryptanalytic attacks, reputation-based attacks, and service-based attacks. Recent blockchain systems are designed with these attacks in mind, and most of the attacks can be defeated if the system is configured following best practices. The design of permissioned consortium blockchains inherently makes many attacks more difficult. For instance, most blockchains use unique transaction IDs and nonces to protect transactions from replay attacks. Permissioned consortium blockchains make Sybil attacks difficult to conduct as identity management is limited to organization administrators.

Regarding DISP, its blockchain security lies within the design and implementation of Hyperledger Fabric. Brotsis et al. [53] highlighted four attack surfaces of Hyperledger Fabric, namely consensus, chaincode, network, and privacy-preserving mechanisms. Hyperledger Fabric is protected against most consensus-oriented attacks but it is more vulnerable to non-deterministic behaviors in chaincode implementation and compromised participants. While the latter type of threat can be eased with careful deployment and maintenance, meticulous attention has been given to the design and implementation of DISP chaincodes to eliminate non-deterministic behavior and ensure the consistency of transactional data. For example, the chaincodes use deterministic JavaScript Object Notation (JSON) serialization libraries to format results. In addition, the application generates all timestamps in the transactions instead of creating them when chaincodes are executing on peers. By doing so, we can eliminate failed transactions due to the system clock not synchronizing across peer nodes. Finally, chaincodes always check the caller's identity and input parameters to prevent impersonation attacks and invalid requests.

For privacy, the identities and transactions are visible to all consortium participants. Although this is usually expected in a trusted environment, users of DISP have the option to conceal the IoT data with the help of auxiliary storage and encryption. Moreover, the private data collection feature offered by Hyperledger Fabric and zero-knowledge proofs [54] are promising approaches that can improve identity and data privacy in our platform. We leave this as future work.

(2) Auxiliary storage security and privacy. We discuss the security and privacy of the auxiliary storage systems using the confidentiality, integrity, and availability (CIA) model. Confidentiality means that the IoT data in the storage should be accessible only to authorized users. In DISP, securely passing sensitive data between a service provider and a consumer can be achieved using a one-time encryption key or access token in terms of data streams. A key may be asymmetrically encrypted using the receiver's identity and sent via the blockchain. Regarding confidentiality during data transfer, the blockchain and auxiliary storage enforce encryption through Transport Layer Security (TLS). Integrity ensures data authenticity, i.e., no unauthorized alteration to data. The data may be modified accidentally due to system errors or by a malicious party. DISP utilizes blockchain as a layer of data integrity assurance for data in auxiliary storage by virtue of its immutability. Therefore, data providers are encouraged to include a digital digest and signature of the

data alongside URI in the service requests and responses. Finally, availability measures how often the data are accessible to its users. For IoT data that desire a high level of availability, distributed storage schemes such as IPFS may be used to facilitate data dissemination and improve data availability. DISP consortium administrators and application developers have the flexibility of realizing data availability.

(3) *IoT device security and privacy.* IoT devices have a long history of being a weak link to IoT system security. Apart from being exposed to physical attacks such as node capturing, sleep deprivation, and false-data injection [55], IoT devices are also vulnerable to network-based attacks. For example, insecurely configured devices are often targeted by IoT botnets [45]. An attacker can acquire access to such devices by brute force attacks or exploiting software flaws. Once successful, malware will be injected into these devices to grow the zombie network or initiate DDoS attacks against other targets.

The proposed DISP can remedy network-based attacks against IoT devices. It offers a secure communication channel to IoT networks that can replace insecure communication protocols such as Telnet and HTTP. The attack surface of IoT devices shrinks as the number of needed services decreases. DISP also eliminates the need for weak credentials by employing strong cryptographic keys and certificates. It also enables the automatic decommission of old IoT devices or decommission of compromised devices using certificate revocation mechanisms. Finally, the use of blockchain and decentralized storage also enhances system security due to the absence of a centralized server, which is often the SPOF in IoT systems.

6. Conclusions

This paper presented an innovative platform called DISP for secure and decentralized IoT communications utilizing the consortium blockchain. DISP models IoT communications as services supported by smart contracts. The service provider, usually an IoT device, exchanges messages with service users securely through blockchain transactions. To support a wide range of applications, DISP also incorporates an auxiliary storage system as a secondary communication channel whose data integrity can be assured by the blockchain. Meanwhile, the inclusion of platform SDKs and gateway makes it easy to integrate the proposed platform into existing IoT systems and devices. Furthermore, a prototype implementation as well as exemplary applications are presented to showcase DISP's generality and versatility. This paper also elaborates on the experimental setup, methodology, and metrics we used to evaluate the performance of DISP. Since the performance of a blockchain system is influenced by a variety of factors, we measure the platform's transaction throughput, latency, and hardware resource utilization under different blockchain configurations and connection sizes. The results indicate that the performance of read operations primarily depends on message size, while the transaction operation performance is subject to batch size, batch timeout, and connection size. Our proof-of-concept implementation can achieve a throughput of 800 RPS and latency of 50ms for read transactions, and a throughput of 80 TPS and latency of 1s for write transactions when the blockchain parameters are optimized. Overall, our proposed work shows great performance and usability potential as a blockchain-based secure communication platform for IoT.

For future work, one direction will be focused on improving transaction throughput and latency for transaction operations on the platform using state-of-the-art lightweight consensus algorithms. Additionally, we plan to investigate new approaches that integrate auxiliary storage with the blockchain to provide the same level of data security and integrity as blockchain transactions. Finally, we will explore new ideas to address privacy concerns and support private services.

Author Contributions: Conceptualization, R.Z. and M.X.; methodology, R.Z.; software, R.Z.; validation, R.Z. and C.X.; formal analysis, R.Z.; investigation, R.Z.; resources, R.Z. and M.X.; data curation, R.Z. and C.X.; writing—original draft preparation, R.Z.; writing—review and editing, M.X., C.X., and R.Z.; visualization, R.Z.; supervision, M.X.; project administration, M.X.; funding acquisition, M.X. All authors have read and agreed to the published version of the manuscript. Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Data presented in this study are available at https://github.com/ nexus-lab/?q=iot-service-blockchain (accessed on 22 September 2022).

Acknowledgments: The authors thank the University of Tennessee at Chattanooga for its support for this study and the Chameleon Cloud for providing experiment resources to this study.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

- attribute-based access control ABAC
- ACL access control list API application programming interface CA certificate authority CBAC capability-based access control CIA confidentiality, integrity, and availability CRL certificate revocation list DDoS distributed denial-of-service H2M human-to-machine ICS industrial control system IdM identity management IoT Internet of Things IPFS InterPlanetary File System **JSON** JavaScript object notation M2M machine-to-machine MSP membership service provider PBFT practical byzantine fault tolerance PM particulate matter RBAC role-based access control RPS read per second RTT round-trip time SDK software development kit SPOF single point of failure SUT system under test TLS transport layer security TPS transaction per second URI UNIFORM RESOURCE IDENTIFIER
- UUID universally unique identifier

References

- Zhang, H.; Sakurai, K. Blockchain for IoT-Based Digital Supply Chain: A Survey. In Proceedings of the Advances in Internet, Data 1. and Web Technologies, Kitakyushu, Japan, 24–26 February 2020; Barolli, L., Okada, Y., Amato, F., Eds.; Springer International Publishing: Cham, Switzerland, 2020; pp. 564–573. [CrossRef]
- 2. Musleh, A.S.; Yao, G.; Muyeen, S.M. Blockchain Applications in Smart Grid–Review and Frameworks. IEEE Access 2019, 7,86746-86757. [CrossRef]
- Ray, P.P.; Dash, D.; Salah, K.; Kumar, N. Blockchain for IoT-Based Healthcare: Background, Consensus, Platforms, and Use Cases. 3. IEEE Syst. J. 2021, 15, 85-94. [CrossRef]
- AbuNaser, M.; Alkhatib, A.A. Advanced survey of Blockchain for the Internet of Things Smart Home. In Proceedings of the 4. 2019 IEEE Jordan International Joint Conference on Electrical Engineering and Information Technology (JEEIT), Amman, Jordan, 9-11 April 2019; pp. 58-62. [CrossRef]
- 5. Henry, R.; Herzberg, A.; Kate, A. Blockchain Access Privacy: Challenges and Directions. IEEE Secur. Priv. 2018, 16, 38-45. [CrossRef]
- 6. Salimitari, M.; Chatterjee, M. A Survey on Consensus Protocols in Blockchain for IoT Networks arXiv 2018. [CrossRef]

- Dib, O.; Brousmiche, K.L.; Durand, A.; Thea, E.; Hamida, E.B. Consortium Blockchains: Overview, Applications and Challenges. *Int. J. Adv. Telecommun.* 2018, 11, 51–64.
- Castro, M.; Liskov, B. Practical Byzantine Fault Tolerance. In Proceedings of the Third Symposium on Operating Systems Design and Implementation, OSDI '99, New Orleans, LA, USA, 22–25 February 1999; USENIX Association: Berkeley, CA, USA, 1999; pp. 173–186.
- Ongaro, D.; Ousterhout, J. In Search of an Understandable Consensus Algorithm. In Proceedings of the 2014 USENIX Annual Technical Conference (USENIX ATC 14), Philadelphia, PA, USA, 19–20 June 2014; USENIX Association: Berkeley, CA, USA, 2014; pp. 305–319.
- 10. Bamakan, S.M.H.; Motavali, A.; Babaei Bondarti, A. A survey of blockchain consensus algorithms performance evaluation criteria. *Expert Syst. Appl.* **2020**, *154*, 113385. [CrossRef]
- 11. Ali, J.; Ali, T.; Musa, S.; Zahrani, A. Towards Secure IoT Communication with Smart Contracts in a Blockchain Infrastructure. *Int. J. Adv. Comput. Sci. Appl.* **2018**, *9*, 578–585. [CrossRef]
- 12. Hang, L.; Kim, D.H. Design and Implementation of an Integrated IoT Blockchain Platform for Sensing Data Integrity. *Sensors* **2019**, *19*, 2228. [CrossRef] [PubMed]
- Zhang, J.; Lu, C.; Cheng, G.; Guo, T.; Kang, J.; Zhang, X.; Yuan, X.; Yan, X. A Blockchain-Based Trusted Edge Platform in Edge Computing Environment. *Sensors* 2021, 21, 2126. [CrossRef] [PubMed]
- Panarello, A.; Tapas, N.; Merlino, G.; Longo, F.; Puliafito, A. Blockchain and IoT Integration: A Systematic Survey. Sensors 2018, 18, 2575. [CrossRef] [PubMed]
- 15. Zhu, X.; Badr, Y. Identity Management Systems for the Internet of Things: A Survey Towards Blockchain Solutions. *Sensors* **2018**, *18*, 4215. [CrossRef] [PubMed]
- Fromknecht, C.; Velicanu, D.; Yakoubov, S. CertCoin: A NameCoin Based Decentralized Authentication System 6.857 Class Project. Available online: https://courses.csail.mit.edu/6.857/2014/files/19-fromknecht-velicann-yakoubov-certcoin.pdf (accessed on 22 September 2022).
- 17. Axon, L. Privacy-Awareness in Blockchain-Based PKI. Available online: https://ora.ox.ac.uk/objects/uuid:f8377b69-599b-4cae-8df0-f0cded53e63b/datastreams/ATTACHMENT01 (accessed on 22 September 2022).
- Al-Bassam, M. SCPKI: A Smart Contract-Based PKI and Identity System. In Proceedings of the ACM Workshop on Blockchain, Cryptocurrencies and Contracts, BCC '17, Abu Dhabi, United Arab Emirates, 2 April 2017; Association for Computing Machinery: New York, NY, USA, 2017; pp. 35–40. [CrossRef]
- Bouras, M.A.; Lu, Q.; Dhelim, S.; Ning, H. A Lightweight Blockchain-Based IoT Identity Management Approach. *Future Internet* 2021, 13, 24. [CrossRef]
- Reed, D.; Law, J.; Hardman, D. The Technical Foundations of Sovrin. Available online: https://sovrin.org/wp-content/uploads/ 2017/04/The-Technical-Foundations-of-Sovrin.pdf (accessed on 22 September 2022).
- 21. Hyperledger Indy. Available online: https://www.hyperledger.org/use/hyperledger-indy (accessed on 21 October 2021).
- Androulaki, E.; Barger, A.; Bortnikov, V.; Cachin, C.; Christidis, K.; De Caro, A.; Enyeart, D.; Ferris, C.; Laventman, G.; Manevich, Y.; et al. Hyperledger Fabric: A Distributed Operating System for Permissioned Blockchains. In Proceedings of the Thirteenth EuroSys Conference, EuroSys '18, Porto, Portugal, 23–26 April 2018; Association for Computing Machinery: New York, NY, USA, 2018. [CrossRef]
- Riabi, I.; Ayed, H.K.B.; Saidane, L.A. A survey on Blockchain based access control for Internet of Things. In Proceedings of the 2019 15th International Wireless Communications & Mobile Computing Conference (IWCMC), Tangier, Morocco, 24–28 June 2019; pp. 502–507. [CrossRef]
- 24. Ouaddah, A.; Abou Elkalam, A.; Ait Ouahman, A. FairAccess: A new Blockchain-based access control framework for the Internet of Things. *Secur. Commun. Networks* 2016, *9*, 5943–5964. [CrossRef]
- Alphand, O.; Amoretti, M.; Claeys, T.; Dall'Asta, S.; Duda, A.; Ferrari, G.; Rousseau, F.; Tourancheau, B.; Veltri, L.; Zanichelli, F. IoTChain: A Blockchain Security Architecture for the Internet of Things. In Proceedings of the 2018 IEEE Wireless Communications and Networking Conference (WCNC), Barcelona, Spain, 15–18 April 2018; pp. 1–6. [CrossRef]
- Bouras, M.A.; Xia, B.; Abuassba, A.O.; Ning, H.; Lu, Q. IoT-CCAC: A blockchain-based consortium capability access control approach for IoT. *PeerJ Comput. Sci.* 2021, 7, e455. [CrossRef] [PubMed]
- Benisi, N.Z.; Aminian, M.; Javadi, B. Blockchain-based decentralized storage networks: A survey. J. Netw. Comput. Appl. 2020, 162, 102656. [CrossRef]
- Storj: A Decentralized Cloud Storage Network Framework. Available online: https://www.storj.io/storj.pdf (accessed on 30 August 2022).
- Vorick, D.; Champine, L. Sia: Simple Decentralized Storage. Available online: https://blockchainlab.com/pdf/whitepaper3.pdf (accessed on 30 August 2022).
- A Decentralized Storage Network for Humanity's Most Important Information. Filecoin. Available online: https://fileCoin.io/(accessed on 21 October 2021).
- Özyilmaz, K.R.; Doğan, M.; Yurdakul, A. IDMoB: IoT Data Marketplace on Blockchain. In Proceedings of the 2018 Crypto Valley Conference on Blockchain Technology (CVCBT), Zug, Switzerland, 20–22 June 2018; pp. 11–19. [CrossRef]

- 32. Tzianos, P.; Pipelidis, G.; Tsiamitros, N. Hermes: An Open and Transparent Marketplace for IoT Sensor Data over Distributed Ledgers. In Proceedings of the 2019 IEEE International Conference on Blockchain and Cryptocurrency (ICBC), Seoul, Korea, 14–17 May 2019; pp. 167–170. [CrossRef]
- Bajoudah, S.; Dong, C.; Missier, P. Toward a Decentralized, Trust-less Marketplace for Brokered IoT Data Trading using Blockchain. In Proceedings of the 2019 IEEE International Conference on Blockchain (Blockchain), Atlanta, GA, USA, 14–17 July 2019; pp. 339–346. [CrossRef]
- Lin, C.H.V.; Huang, C.C.J.; Yuan, Y.H.; Yuan, Z.s.S. A Fully Decentralized Infrastructure for Subscription-based IoT Data Trading. In Proceedings of the 2020 IEEE International Conference on Blockchain (Blockchain), Rhodes, Greece, 2–6 November 2020; pp. 162–169. [CrossRef]
- 35. Home | IOTA. Available online: https://data.iota.org/ (accessed on 21 October 2021).
- 36. XBR: Open Data Markets Infrastructure. Available online: https://xbr.network/ (accessed on 21 October 2021).
- 37. Unstoppable Data for Unstoppable Apps: DATAcoin: By Streamr. Available online: https://www.securities.io/wp-content/uploads/2022/04/streamr-datacoin-whitepaper-2017-07-25-v1_1.pdf (accessed on 22 September 2022).
- 38. Slock.it. Available online: https://web.archive.org/web/20150724122424/http://slock.it/ (accessed on 21 October 2021).
- Fakhri, D.; Mutijarsa, K. Secure IoT Communication using Blockchain Technology. In Proceedings of the 2018 International Symposium on Electronics and Smart Devices (ISESD), Bandung, Indonesia, 23–24 October 2018; pp. 1–6. [CrossRef]
- 40. MQTT—The Standard for IoT Messaging. Available online: https://mqtt.org (accessed on 22 September 2022).
- Wickström, J.; Westerlund, M.; Pulkkis, G. Smart Contract based Distributed IoT Security: A Protocol for Autonomous Device Management. In Proceedings of the 2021 IEEE/ACM 21st International Symposium on Cluster, Cloud and Internet Computing (CCGrid), Melbourne, Australia, 10–13 May 2021, pp. 776–781. [CrossRef]
- Wake Word Detection & Keyword Spotting | Picovoice On-Device Voice AI. Available online: https://picovoice.ai/platform/ porcupine/ (accessed on 14 May 2022).
- Baabaaox/Go-Webrtcvad: WebRTC Voice Activity Detection for Golang. Available online: https://github.com/baabaaox/gowebrtcvad (accessed on 14 May 2022).
- Voice Commands & NLU | Picovoice On-Device Voice Recognition. Available online: https://picovoice.ai/platform/rhino/ (accessed on 14 May 2022).
- Antonakakis, M.; April, T.; Bailey, M.; Bernhard, M.; Bursztein, E.; Cochran, J.; Durumeric, Z.; Halderman, J.A.; Invernizzi, L.; Kallitsis, M.; et al. Understanding the Mirai Botnet. In Proceedings of the 26th USENIX Security Symposium (USENIX Security 17), Vancouver, BC, Canada, 3 May 2017; USENIX Association: Berkeley, CA, USA, 2017; pp. 1093–1110.
- Keahey, K.; Mambretti, J.; Ruth, P.; Stanzione, D. Chameleon: A Large-Scale, Deeply Reconfigurable Testbed for Computer Science Research. In Proceedings of the 2019 IEEE 27th International Conference on Network Protocols (ICNP), Chicago, IL, USA, 8–10 October 2019; pp. 1–2. [CrossRef]
- 47. Hyperledger Caliper. Available online: https://hyperledger.github.io/caliper/ (accessed on 5 May 2022).
- Xu, L.; Chen, W.; Li, Z.; Xu, J.; Liu, A.; Zhao, L. Solutions for concurrency conflict problem on Hyperledger Fabric. World Wide Web 2021, 24, 463–482. [CrossRef]
- Performance, H.; Group, S.W. Hyperledger Blockchain Performance Metrics. Available online: https://www.hyperledger.org/learn/publications/blockchain-performance-metrics (accessed on 27 March 2022).
- 50. Prometheus—Monitoring System & Time Series Database. Available online: https://prometheus.io (accessed on 22 September 2022).
- Baliga, A.; Solanki, N.; Verekar, S.; Pednekar, A.; Kamat, P.; Chatterjee, S. Performance Characterization of Hyperledger Fabric. In Proceedings of the 2018 Crypto Valley Conference on Blockchain Technology (CVCBT), Zug, Switzerland, 20–22 June 2018; pp. 65–74. [CrossRef]
- Ferrag, M.A.; Derdour, M.; Mukherjee, M.; Derhab, A.; Maglaras, L.; Janicke, H. Blockchain Technologies for the Internet of Things: Research Issues and Challenges. *IEEE Internet Things J.* 2019, *6*, 2188–2204. [CrossRef]
- Brotsis, S.; Kolokotronis, N.; Limniotis, K.; Bendiab, G.; Shiaeles, S. On the Security and Privacy of Hyperledger Fabric: Challenges and Open Issues. In Proceedings of the 2020 IEEE World Congress on Services (SERVICES), Beijing, China, 18–23 October 2020; pp. 197–204. [CrossRef]
- Kang, H.; Dai, T.; Jean-Louis, N.; Tao, S.; Gu, X. FabZK: Supporting Privacy-Preserving, Auditable Smart Contracts in Hyperledger Fabric. In Proceedings of the 2019 49th Annual IEEE/IFIP International Conference on Dependable Systems and Networks (DSN), Portland, OR, USA, 24–27 June 2019; pp. 543–555. [CrossRef]
- Hassija, V.; Chamola, V.; Saxena, V.; Jain, D.; Goyal, P.; Sikdar, B. A Survey on IoT Security: Application Areas, Security Threats, and Solution Architectures. *IEEE Access* 2019, 7, 82721–82743. [CrossRef]